Author’s response to reviews

Title: Atrophic inferior vena cava is a marker of chronicity of intra-filter and inferior vena cava thrombosis: based on CT findings

Authors:
Liang Chen (Huly_nj@163.com)
Wanyin Shi (shwy110@163.com)
Jianping Gu (gjp_nj@163.com)
Xu He (hexu_nj@163.com)
Wensheng Lou (louwensheng_nj@163.com)

Version: 2 Date: 08 Oct 2017

Author’s response to reviews:

Editor's comments:

1. We have noted that you have now removed your Consent to publish section. If identifying images or other personal or clinical details of participants are presented that compromise anonymity, a statement of consent to publish from the patient should be included. This section must be included even if it is not applicable to your manuscript. If consent to publish is not applicable to your manuscript please write ‘Not Applicable’ in this section.

Please clarify whether the data used in this study was anonymised. If it was not, did the patients know that their data was being used.

Response: Thanks. It is "Not Applicable" to get consent to publish from patients or their immediate family members. The data used in this study was anonymised, including information of name, age, gender that was removed from the abstract, MS, figures, tables, and figure legends.
2. There are many indirect identifiers in Table 1, such as the ages and genders of the patients, please clarify whether the patients or their next of kin consented to this data being published in the Consent to publish section, or remove the identifiers by presenting the ages as ranges and removing gender information.

Response: Thanks, I removed the age and gender information from Table 1.

3. Please ensure that the Abstract entered into Editorial Manager is the same as the one in the main manuscript.

Response: Thanks, I amended it as required.

4. Please remove the "Purpose" heading from the Abstract.

Response: Thanks, I removed it as required.

5. At this stage, we ask that you submit a clean version of your manuscript and do not include track changes or highlighting.

Response: Thanks, I amended it as required.